\clearpage
\addcontentsline{toc}{chapter}{Bibliography}
\begin{thebibliography}{99}

\bibitem{Baade}Baade, W. \& Zwicky, F., 1934, Phys. Rev., 46, 76.



\bibitem{aharonian2004a}Aharonian, F., A., et al. 2004,
%On the non-thermal high energy radiation of galaxy clusters
A\&A 417, 391.

\bibitem{aharonian2005b}Aharonian F., A., et al., A\&A 432, L25.XS

\bibitem{Winkler}Winkler, C., et al. 2003,
%The INTEGRAL mission
A\&A 411, L1-L6.

\bibitem{Johnson}Johnson, W. N., et al. 1993, ApJS, 86, 693

\bibitem{Diehl}Diehl, R., 1989,
%The comptel experiment on the NASA Gamma-Ray Observatory 
%Space Science Reviews, 49 1-2:85-106.
SSR, 49 1--2, 85.

\bibitem{Kanbach}Kanbach, G., et al. 1989,
%The project EGRET (energetic gamma-ray experiment telescope) 
%on NASA's Gamma-Ray Observatory GRO
%Space Science Reviews, Volume 49, Numbers 1-2:69:84. 
SSR, Volume 49, Numbers 1--2, 69. 

%\cite{Gehrels:1999ri}
\bibitem{Gehrels:1999ri}Gehrels, N., Michelson, P., 1999,
 % N.~Gehrels and P.~Michelson,
 %GLAST: The next-generation high energy gamma-ray astronomy mission
%Astropart. Phys. 11, 277.
Astropart. Phys. 11, 277.

%\cite{CarterLewis:1997nq}
\bibitem{CarterLewis:1997nq}
 Carter-Lewis, D.~A. et al., 1997,
% Spectrum of TeV gamma rays from the Crab Nebula
arXiv:astro-ph/9706145.
  %%CITATION = ASTRO-PH/9706145;%%

%\cite{Karle:1995dk}
\bibitem{Karle:1995dk}Karle, A. 1995
  %``Design and performance of the angle integrating Cherenkov 
%array AIROBICC,''
  Astropart.\ Phys.\ 3, 321.
  %%CITATION = APHYE,3,321;%%

%\cite{Sinnis:1994iy}
\bibitem{Sinnis:1994iy}Sinnis, G., [Milagro Collaboration], 1995
  %``Milagro: A low energy threshold extensive air shower array,''
  Nucl.\ Phys.\ Proc.\ Suppl.\ 43, 141.
  %%CITATION = NUPHZ,43,141;%%

\bibitem{greisen}Greisen, K. 1966, Phys. Rev. Lett. 16, 748.

\bibitem{halzen}Halzen, F. \& Hooper, D. 2002, Rept. Prog. Phys, 65, 1025.

\bibitem{kappes}Kappes, A. et al, 2006, astro--ph/0607286.

\bibitem{Stecker} Stecker, F.W., 1971, Cosmic Gamma Ray (Baltimore: Mono).

\bibitem{Stecker1992}Stecker, F. W., de Jager, O. C., Salamon, M. H. 1992, ApJ, 390, L49.

\bibitem{blatnig} Blatnig, S.R. er al., NASA/TP--2000--210640.

\bibitem{domingo}Domingo--Santamaria, E. \& Torres, D. F., 2005, A\&A,
  444, 403.

\bibitem{Aharonian2000}Aharonian, F. A. \& Atoyan, A. M., 2000. A\&A,
  362, 937.

\bibitem{AharonianAtoyan1997}Aharonian, F. A.,  Atoyan, A. M., \&
  Kifune,  T., 1997, MNRAS, 291, 162.

\bibitem{gaisser1990}Gaisser, T. K., 1990, Cosmic Rays and Particle
  physics, Cambridge University Press.

\bibitem{woltjer}Woltjer, L., 1972, ARA\&A, 10, 129.

\bibitem{berezhko}Berezhko, E. G. \& V\"{o}lk, H. J., 1997,
  Asropart. Phys., 7, 183.

\bibitem{rees1967}Rees, M. J., 1967, MNRAS, 137, 429.

\bibitem{blumental}Blumenthal, G. R., 1970, Rev. Mod. Phys., 42, 237.

\bibitem{schlickeiser}Schlickeiser, R., 2002, 
Cosmic Ray Astrphysics, Springer-Verlag, New York.

\bibitem{jackson1975}Jackson, J. D., 1975, 
Classical Electrodynamics, Wiley, New York.

\bibitem{bell}Bell, A. R., 1978, MNRAS, 1982, 147.

\bibitem{heitler}Heitler, W., 1954, 
The Quantum Theory of Radiation, Oxford, Clarendon Press.

\bibitem{uchiyama}Uchiyama, Y., et al., 2002, ApJ, 571, 866.

\bibitem{zhang}Zhang, J. L. \& Yuan, Y. F., 1998, ApJ, 439, 826.

\bibitem{lovelace}Lovelace, R. V. E.,1976, Nature, 262, 649.

\bibitem{jungman}Jungman, G., Kamionkowsky, M. \& Griest, K., 1996, 
Phys Rep., 267, 195.

%%%%%%%%%%%%%%%%%%%%%%%%%%
\bibitem{ellis}Ellis J. R. et al., 1984. Nucl. Phys., B238, 453.

\bibitem{bergstrom}Bergstrom, L., Ullio, P. \& Buckley, J. H., 1998, 
Astropart. Phys., 9, 137.

%\cite{Krawczynski2004}
\bibitem{Krawczynski2004}Krawczynski, H. et al., 2004, ApJ, 601, 151.

%\cite{Cronin}
\bibitem{Cronin}Cronin, J., Gaisser, T. K. \& Swordy, S. P., 1997, 
Sci. Amer., 276, 44.

%\cite{Kampert}
\bibitem{Kampert}Kampert, K. H., (Auger Collab.), 2006, astro--ph/0608136.

%\cite{hess1912}
\bibitem{hess1912}Hess, V., 1912, Physik. Zeitschr., 13, 1084.

%\cite{Yao}
\bibitem{Yao}Yao, W. M., et. al., 2006, J. Phys., G33, 1.

%\cite{Hillas2005}
\bibitem{Hillas2005}Hillas A. M., 2005, J. Phys., G31, 95.

%\cite{Haungs2003}
\bibitem{Haungs2003}Haungs, A., 2003, J. Phys. G. 29, 809.

%\cite{Ginzburg}
\bibitem{Ginzburg}Ginzburg V. L., \& Syrovatskii, S. I., 1964, 
The Origin of Cosmic Rays, New York: MacMillian.

\bibitem{AharonianDruryVoelk}Aharonian, F. A., Drury, L. O'C \& Voelk, H. J., 1994, A\&A, 285, 645.

%\cite{Gaisser2001}
\bibitem{Gaisser2001}Gaisser T. K., 2001 AIP Conf. Proc., 558, 27.

\bibitem{Swordy1990}Swordy, S. P., et al., 1990, ApJ, 349, 625.

\bibitem{SeoPtuskin1994}Seo, E. S.\& Ptuskin, V. S., 1994, ApJ. 431, 705.

\bibitem{Longair1994}Longair, M. S., 1994, 
High Energy astrophysics, Cambridge University Press.

\bibitem{MalkovDrury}Malkov, M. A.,O'C Drury, L., 2001, Rep. Prog. Phys., 64, 429.

\bibitem{fermi1954}Fermi, E., 1954, ApJ, 119, 1.

\bibitem{BlandfordEichler1987}Blandford, R.\& Eichler, D., 1987, Phys. Rep., 150, 1.

\bibitem{Arons}Arons, J., 1998, Mem, Soc. Ast. Ital., 69, 989.

\bibitem{Rudak2001}Rudak, B., 2001 astro--ph/0101138

\bibitem{LyubarskyKirk2001}Lyubarsky, Y. \& Kirk, J.G., 2001, ApJ, 547, 437.

\bibitem{Hurt}Hurt, R., 2005, SSC/Caltech.

\bibitem{CordesLazio2002}Cordes, J. M. \& LazioT. J. W., 2002, astro--ph/0207156.

\bibitem{Schodel}Sch\"odel, R. et al., 2002, Nature, 419, 694.

\bibitem{Richstone}Richstone, D et al., 1998, Nature, 395, A14.

\bibitem{Battaglia}Battaglia, G. et al. 2005, MNRAS, 364, 433.

\bibitem{CordesLazio}Cordes, J. M. \& Lazio, T. J. W., 2002, astro-ph/0207156.

\bibitem{Nomoto}Nomoto, K., Thielemann, F. K., Yokoi, K., 1984, ApJ, 286, 644.

\bibitem{Bethe}Bethe, H. A., 1990, Rev. Mod. Phys., 62, 801.

\bibitem{Woltjer}Woltjer, L., , 1972, ARA\&A, 10, 129.

\bibitem{Woosley}Woosley, S. E. \& Weaver, T. A., 1995 ApJS, 101, 181.

\bibitem{Koyama}Koyama, K., 1995, Nature, 378, 255.

\bibitem{Enomoto2002}Enomoto, R. et al (CANGAROO Collab.), 
2002, Nature, 416, 823.

\bibitem{Aharonian2004c}Aharonian, F. A. et al. (HESS Collab.), 
2004c, Nature, 432, 75.

\bibitem{Goldreich}Goldreich, P. \& Julian, W. H., 1969, ApJ, 157, 869.

\bibitem{Harding1981}Harding, A. K., 1981, ApJ, 245, 267.

\bibitem{Gaensler2000}Gaensler, B. M., 2000, MNRAS, 317, 58.

\bibitem{Jager1992} de Jager, O. C., Harding, A. K., 1992, ApJ, 396, 161.

\bibitem{Harding1996}Harding, A. K., 1996, Space Sci. Rev., 75, 257.

\bibitem{Safi-Harb} Safi-Harb, S., 2004, BAAA, 47, 277.

\bibitem{Weekes1998}Weekes, T. C. (Whipple Collab.), 1998, ApJ, 503, 744.

\bibitem{Aharonian2005d}Aharonian, F. A. et al. (HESS Collab.), 
2005, A\&A, 442, 1.
\bibitem{Aharonian2005e}Aharonian, F. A. et al. (HESS Collab.), 
2005, Science 309, 746.

\bibitem{Albert2006d}Albert, J. et al. (MAGIC Collab.), 2006, 
Science 312, 1771.

\bibitem{Mirabel2006}Mirabel, I. F., 2006, Science, 312, 1759.

\bibitem{Paredes}Paredes, J. M., 2005, AIPC, 745, 93.

\bibitem{Mirabel1992}Mirabel, I. F. et al., 1992, Nature, 358, 215.

\bibitem{Heinz}Heinz, S. and Sunyaev, R. A., 2002, A\&A, 390.

\bibitem{Dyson}Dyson, J. 1997, 
Physics of the Interstellar Medium, London: Taylor \& Francis.

\bibitem{Jackson2006}Jackson, J. M., et al. 2006, ApJS, 163, 145.

\bibitem{Torres2003}Torres, D. F. et al., 2003, Phys. Rept., 382, 303.

%\bibitem{Aharonian2006c}Aharonian, F. A., et al. (HESS Collab.), 2006c, 
%Nature, 439, 695.

%\bibitem{AharonianNeronov}Aharonian, F. A. Neronov, A., 2005, ApJ, 619, 306.

\bibitem{Horns}Horns, D., 2005, Phys. Lett., B607, 225.

\bibitem{Wang}Wang, Q. D. et al., 2006, MNRAS, 367, 937.

\bibitem{Eisenhauer2005}Eisenhauer, F., et al., 2005, ApJ, 628, 246.

\bibitem{Bower2004}Bower, C. B. et al., 2004, Science, 304, 704.

\bibitem{Mayer-Hasselwander}Mayer-Hasselwander, H. A. et al., 1998, A\&A, 335, 161.

\bibitem{HooperDingus}Hooper, D. \& Dingus, B., 2005, AdSpR, 35, 130, astro-ph/0212509.

\bibitem{Tsuchiya2004}Tsuchiya, K. et al. (CAnGAROO Collab.), 2004, ApJ, 606, L115.

\bibitem{Kosack}Kosack, K. et al., 2004, ApJ, 608, L97.

\bibitem{Aharonian2004b}Aharonian, F. A. et al. (HESS Collab.), 2004, A\&A, 425, L13.

\bibitem{Aharonian2005b}Aharonian, F. A. et al. (HESS Collab.), 2005b, A\&A, 432, L25.

\bibitem{FalckeMarkoff}Falcke, H. \& Markoff S. 2000, A\&A, 362, 113.

\bibitem{FatuzzoMelia}Fatuzzo, M. \& Melia, F., 2003, ApJ, 596, 1035.

\bibitem{Wang2006}Wang, Q. D. et al., 2006, MNRAS, 367, 937.

\bibitem{Aharonian2006c}Aharonian, F. A., et al. (HESS Collab.), 2006c, Nature, 439, 695.

\bibitem{Pohl1997}Pohl, M., 1997, A\&A, 317, 441.

\bibitem{Prada2004}Prada, F. et al., 2004, Phys. Rev. Lett., 93, 241301.

\bibitem{Maeda2002}Maeda, Y. et al., 2002, ApJ, 570, 671.

\bibitem{Yusef-Zadeh1999}Yusef-Zadeh, F. et al., 1999, ApJ, 527, 172.

\bibitem{AharonianNeronov}Aharonian, F. A. \& Neronov, A., 2005, ApJ, 619, 306.

\bibitem{AtoyanDermer}Atoyan, A. \& Dermer, C. D., 2004, ApJ, 617, L123.

\bibitem{Horns2005}Horns, D., 2005, Phys. Lett., B607, 225.

\bibitem{Hayashida1999}Hayashida, N. et al. (AGASA Collab.), 1999, Astropart. Phys., 10, 303.

\bibitem{Haslam}Haslam, C. G. T., 1982, A\&AS, 47, 1.

\bibitem{Dame2001}Dame, T. M., Hartmann, D. \& Thaddeus, P., 2001, ApJ, 547, 792.

\bibitem{Snowden}Snowden, S. L. et al., 1997, ApJ, 485, 125.

\bibitem{Hartman}Hartman, R. C. et al., 1999, ApJS, 123, 79.

\bibitem{Aharonian2006a}Aharonian, F. A. et al. (HESS Collab.), 2006a, ApJ, 636, 777.

\bibitem{Goodman}Goodman, J. A. et al., 2006, 
Procs. 2nd Workshop On TeV Particle Astrophysics, Madison, WI, USA.

\bibitem{Aharonian2005a}Aharonian, F. A. et al. (HESS Collab.), 2005a, Science, 307, 1938.

\bibitem{Atkins2004}Atkins, R. et al., 2004, ApJ, 608, 680.

\bibitem{Amenomori2002}Amenomori, M. et al, 2002, ApJ, 580, 887.

\bibitem{Atkin2005}Atkins, R. et al., 2005, Phys. Rev. Lett., 95, 1103.

\bibitem{Aharonian2002}Aharonian, F. A. et al. (HEGRA Collab.), 2002, A\&A, 395, 803.

\bibitem{UrryPadovani}Urry, C. \& Padovani, P., 1995, PASP, 107, 803.

\bibitem{Rees1984}Rees, M. J., 1984, ARA\&A, 22, 471.


\bibitem{Punch}Punch, M. et al. (Whipple Collab.), 1992, Nature, 358, 477.

\bibitem{Aharonian2003}Aharonian, F. A. et al. (HEGRA Collab.), 2003, A\&A, 403, L1.

\bibitem{Biller1999}Biller, S. D. et al., 1999, Phys. Rev. Lett., 83, 2108.

\bibitem{Fossati1998}Fossati, G., et al., 1998, MNRAS, 299, 433.

\bibitem{BlanchMartinez}Blanch, O \& Martinez, M., 2005, Astropart. Phys., 23, 598.

\bibitem{Dwek}Dwek, E. \& Krennrich, F., 2005, ApJ, 618, 657.

\bibitem{Kashlinsky}Kashlinsky, A., 2005, PhR, 409, 361.

\bibitem{HauserDwek}Hauser, M. G.; Dwek, E., 2001, ARA\&A, 39, 249.

\bibitem{Reid}Reid, M. J., 1993, ARA\&A, 31, 345.

\bibitem{Lazio}T. J. W. Lazio, et al., 2006, available at
http ://rsd-www.nrl.navy.mil/7213/lazio/GC/GC Schematicscan2rot.gif.

%\bibitem{}
%%%%%%%%%%%%%%%%%%%%%%%%%%

\%bibitem{Fernow}Fernow, R. C., 1986, 
%Introduction to Experimental Particle Physics, Cambridge University Press.

%\bibitem{Yao}Yao, W.M. et al., 2006, J. Phys., G33, 1.

\bibitem{Eidelman}Eidelman, S. et al. 
(Particle Data Group), 2004, Physics Letters, B592, 1.

\bibitem{Rossi}Rossi, B., 1965, High-Energy Particles, Prentice Hall Series.

\bibitem{Kleinknecht}K. Kleinknecht, 1998, Cambridge Univ. Press, 2nd edition.

\bibitem{Fernow}Fernow, R. C., 1986, 
Introduction to Experimental Particle Physics, Cambridge University Press.

\bibitem{Wigmans}Wigmans, R., 1987, NIM, A259, 389.

\bibitem{Grupen}C. Grupen, A. Bohrer, and L. Smolik, 1996, 
Cambridge Monogr. Part. Phys. Nucl. Phys. Cosmol., 5, 1.


\bibitem{Scott}Scott, W. T., 1963, Rev. Mod. Phys., 35, 231.

\bibitem{Schmidt}Schmidt, F., http://www.ast.leeds.ac.uk/fs/showerimages.html

\bibitem{Leo}Leo, W. R., 1994, 
Techniques for Nuclear and Particle Physics Experiments, Springer.

\bibitem{Barrio}Barrio, J. A. et al. (MAGIC Collab.),  
The MACIC Telescope - Design Study for the Construction of a 17m 
Cherenhov Telescope for Gamma Astronorny above 1O GeV, MPI-PhE-98-05. 

\bibitem{Baixeras2004}Baixeras, C. et al. (MAGIC Collab.), 2004, NIM, A518, 188.

\bibitem{Cortina2005}Cortina, J. et al. (MAGIC Collab.),  2005, Proc. of the 29th ICRC, Pune, India, 5-359, astro-ph/ 0508274 

\bibitem{Kabuki}Kabuki, S. et al. (CANGAROO Collab.), 2003, NIM, A500, 318.

\bibitem{Aharonian2006d}Aharonian, F. A., et al. (HESS Collab.), 2006d, 
A\&A, 457, 899.

\bibitem{Holder}Holder, J. et al. (Veritas Collab.), 2006, Astropart. Phys., 25, 391.

\bibitem{Albert2006c}Albert, J. et al. (MAGIC Collab.), 2006, ApJ, 639, 761.

\bibitem{Ostankov}Ostankov, A. et al., 2000, NIM, A442, 117.

\bibitem{Daum}Daum, A. et al. (HEGRA Collab.), 1997, Astropart. Phys., 8, 1.

\bibitem{Bretz2003}Bretz, T. et al. (MAGIC Collab.), 2003, 
Proc. of the 28th ICRC, Tsukuba, Japan, 2943.

\bibitem{Riegel}Riegel, B. et al. (MAGIC Collab.),  2005, 
Proc. of the 29th ICRC, Pune, India, 5-219, astro-ph/ 0508274.

\bibitem{Bigongiari}Bigongiari, C. et al., 2004, NIM, A518, 193.

\bibitem{Garczarczyk2003}Garczarczyk, M. et al. (MAGIC Collab.) , 2003, 
Proc. of the 28th ICRC, Tsukuba, Japan, 2935.

\bibitem{Cortina2003}Cortina, J. et al. (MAGIC Collab.) , 2003, 
Proc. of the 28th ICRC, Tsukuba, Japan, 2931.

\bibitem{Paneque2004}Paneque, D. et al., 2004, NIM, A 518, 619.

\bibitem{Lorenz2001}Lorenz, E. et al., 2001, NIM, A461, 517.

\bibitem{Gaug2006}Gaug, M., 2006, Ph.D. thesis IFAE Barcelona.

\bibitem{Nakamura}Nakamura et al., 1995, Jap. Journ. of Appl. Phys., 34, L1332.

\bibitem{Gaug2005}Gaug, M. , Bartko, H. et al. (MAGIC Collab.) , 2005,
 Proc. of the 29th ICRC, Pune, India, 5-375 , astro-ph/0508274.

\bibitem{Majumdar}Majumdar, P. et al. (MAGIC Collab.), 2005, 
Proc. of the 29th ICRC, Pune, India, 5-203, astro-ph/0508274.

\bibitem{Heck}Heck, D. et al., 1998, Report FZKA 6019.

\bibitem{Sobczynska}Sobczynska, D., 2002, MAGIC-TDAS 02-10.

\bibitem{Werner}Werner, K., 1993, Phys. Rept., 232, 87.

\bibitem{Kneizys}Kneizys et al., 1996, 
The MODTRAN 2/3 Report and LOWTRAN 7 Model, Phillips
Laboratory, Hanscom AFB, MA 01731, USA.

\bibitem{Moralejo}Moralejo, A., 2002, MAGIC-TDAS 02-05.

\bibitem{Elterman}Elterman, L., 1964, Appl. Opt., 3, 6.

%\bibitem{Blanch}Blanch, O.,  Martinez, M., 2005, Astropart. Phys., 23, 598.

\bibitem{Blanch}Blanch, O., 2003, MAGIC-TDAS 03-06.

\bibitem{Goebel2005}Goebel, F. et al. (MAGIC Collab.),  2005, 
Proc. of the 29th ICRC, Pune, India, 5-375, astro-ph/0508274.

\bibitem{Alcaraz2000a}Alcaraz, J. et al., 2000a, Phys. Lett., 490, 27.

\bibitem{Alcaraz2000b}Alcaraz, J. et al., 2000b, Phys. Lett., 494, 392.

\bibitem{Wittek2001}Wittek, W., 2001, MAGIC--TDAS 01--05.

\bibitem{Fomin}Fomin, V. P. et al., 1994, Astropart. Phys., 2, 137.

\bibitem{Teshima2005}Teshima, M. et al. (MAGIC Collab.), 2005, 
Proc. of the 29th ICRC, Pune, India, 5-227, astro-ph/0508274.

\bibitem{Antoranz}Antoranz, P. et al. (MAGIC Collab.) , 2006, 
Proc. of the ECRC 2006 Lisbon, MPP--2006--143.

\bibitem{Turini}Turini, N. et al. (MAGIC collaboration), 2005, 
Proceedings of the conference Towards a
Networh of Atrnospheric Cherenhov Detectors VII, Palaiseau, France, 493.

%\cite{Paoletti:2007zza}
\bibitem{Paoletti:2007zza}Paoletti, R., et al., 2007,
  %``The trigger system of the MAGIC telescope,''
  IEEE Trans.\ Nucl.\ Sci.\  54, 404.
  %%CITATION = IETNA,54,404;%%

%\cite{Pegna:2006bm}
\bibitem{Pegna:2006bm}Pegna, R., et al., 2006, 
  %``Performance of the Domino Ring Sampler in the MAGIC experiment,''
  NIM A 567, 218.
  %%CITATION = NUIMA,A567,218;%%

%\cite{Meucci:2004jr}
\bibitem{Meucci:2004jr}Meucci, M., [MAGIC Collaboration], 2004,
  %``The trigger system of the MAGIC telescope: On-line 
%selection strategies for
  %Cherenkov telescopes,''
  NIM A 518, 554.
  %%CITATION = NUIMA,A518,554;%%

\bibitem{Bartko}Bartko, H. et al., 2005, NIM, A548, 464.


\bibitem{BretzWagner}Bretz, T. \& R. Wagner (MAGIC Collab.), 2003, 
Proc. of the 28th ICRC, Tsukuba, Japan, 2947.

\bibitem{Press}Press, W. H. et al., 2002, 
Cambridge University Press, second edition.

\bibitem{ClelandStern}Cleland, W. E. \& Stern, E. G., 1994, NIM, A338, 467.

\bibitem{Papoulis}Papoulis, A., 1977, Signal analysis , McGraw-Hill.

\bibitem{Bartko2005a}Bartko, H. et al. (MAGIC Collab.), 2005,
Proceedings of the conference Towards a Network of 
Atmospheric Cherenkov Detectors VII, Palaiseau, France, astro-ph/0506459.

\bibitem{Schweizer}Schweizer, T. et al., 2002, IEEE Trans. Nucl. Sci., 49, 2497.

\bibitem{MirzoyanLorenz1997}Mirzoyan, R. \& Lorenz, E. 1997, Proc. of the 25th ICRC, 
Durban, South Africa, 7, 265.

\bibitem{Tonello}Tonello, N., 2006, Ph.D. thesis, Techical University Munich.

\bibitem{Wittek2002c}Wittek, W., 2002, MAGIC--TDAS 02--14.

\bibitem{Hillas1985}Hillas, A. M., 1985, Proc. of the 19th ICRC, La Jolla, 3, 445.

\bibitem{Fegan}Fegan, D. J., 1997, J. Phys. G., 23, 1013.

%\bibitem{}

%%%%%%%%%%%%%%%%%%%%%%%%%%%

\bibitem{goebel1}F. Goebel et al., 2007,
%Upgrade of the MAGIC 
%Telescope with a Multiplexed Fiber-Optic 2 GSamples/s FADC Data 
%Acquisition System system. 
Proceedings of the XXX International Cosmic Ray Conference.

\bibitem{Mirzoyan1}Mirzoyan, R., et al., 2001,
%Multiple Signal Channel Read Out by a Single FADC. 
International Cosmic Ray Conference, volume 7 of International
Cosmic Ray Conference, 2845.

\bibitem{Mirzoyan2}Mirzoyan r. et al. 2002,
%Ultrafast FADC multiplexer.
Nuclear Science, 29, 2473.

\bibitem{bartko}Bartko, H. et al.,  2005.
%Tests of a prototype multiplexed fiber-optic ultra-fast FADC data 
%acquisition system for the MAGIC telescope.
NIM A 548, 464.

\bibitem{HEGRA} HEGRA Collaboration et al., 1999,
%The time structure of Cherenkov images generated by
%TeV gamma-rays and by cosmic rays. 
Astroparticle Physics, 11, 363.

\bibitem{Tescaro2}Tescaro, D., 
Informazioni temporali nelle immagini cherenkov osservate dal telecopio MAGIC.
Master's thesis, Universit\`a degli Studi di Padova, March 2005

\bibitem{Mirzoyan3}Mirzoyan, R. et al., 2006,
%D. Sobczynska, E. Lorenz, and M. Teshima. 
%Tagging single muons and other long-flying relativistic charged 
%particles by ultra-fast timing in air Cherenkov
%telescopes. 
AP, 25, 342.

\bibitem{Tescaro}Tescaro, D.
Timing analysis of the MAGIC telescope data
after the installation of the ultra-fast 2 GSamples/s
FADC readout, 
PhD Thesis, Institut de F\'isica d'Altes Energies, Universitat Aut\`onoma
de Barcelona, 
September 30, 2007.

\bibitem{CDFnote6259}CDF note 6259

\bibitem{slink}http://hsi.web.cern.ch/HSI/s-link/

\bibitem{scaa032}Texas Instruments
%Minimizing Clock Driver Output Skew Using Ganged Outputs
Application Note scaa032, http://focus.ti.com/lit/an/scaa032/scaa032.pdf

\bibitem{ritt1}Ritt, S. 2007
%Design and performance of the 5 GHz waveform digitizing chip DRS3.
Nuclear Science Symposium Conference Record. NSS '07. IEEE 4, 2485.

\bibitem{drs-ritt-2002}Ritt, S.,  2002,
IEEE Trans. Nucl. Sci. 49, 1177.

\bibitem{Kleinfelder}Kleinfelder, S. 2003
IEEE TRANSACTIONS ON NUCLEAR SCIENCE, VOL. 50, NO. 4.

\bibitem{Albrecht}Albrecht K., 2006,
%Nuclear Instruments and Methods in Physics Research Section A,
%Volume 567, Issue 2, 15 November 2006, Pages 438-443
NIM A, 567, Issue 2, 438.

\bibitem{Delagnes}Delagnes, E., 2006,
%SAM: A new GHz sampling ASIC for the H.E.S.S.-II front-end electronics, 
%Nuclear Instruments and Methods in Physics Research Section A,Volume 567,
% Issue 1, 1 November 2006, Pages 21-26 
NIM A 567, 21.

\bibitem{Varner}Varner, G., 2007, 
%Nuclear Instruments and Methods in Physics Research Section A, Volume 583, Issue 2-3, p. 447-46
NIM A 583, 447.

\bibitem{barcelo_2007}Marcelo, M. et al., 2007,
NIMA 572, 382
% A GHz sampling Readout System as Data Acquisition for the MAGIC-II 
% Telescope*
% M. Barcelo, M. Bitossi, R. Cecchi, M. Fagiolini, R. Paoletti, R. Pegna, 
% A. Piccioli
% *NIMA (2007) 527, 382-384*
% Prepared for the 10th Pisa Meeting on Advanced Detectors, La Biodola, 
% Isola d`Elba, Mar 21-25, 2006 Submitted to Nuclear Instruments and Methods A

\bibitem{Broennimann}Broennimann C., et al., 1999,  NIM A420, 264.

\bibitem{Anikeev}Anikeev K., et al. (CDF Collaboration), 2006,
IEEE Trans. Nucl. Sci. NS-53,  653.

\end{thebibliography}